Author's response to reviews

Title: Association of the sense of coherence with physical and psychosocial health in the rehabilitation of osteoarthritis of the hip and knee: a prospective cohort study.

Authors:

Thomas Benz (t.benz@rehaclinic.ch)
Felix Angst (f.angst@rehaclinic.ch)
Susanne Lehmann (s.lehmann@rehaclinic.ch)
André Aeschlimann (a.aeschlimann@rehaclinic.ch)

Version: 3 Date: 30 April 2013

Author's response to reviews: see over
Author’s response to reviews

Title: Association of the sense of coherence with physical and psychosocial health in the rehabilitation of osteoarthritis of the hip and knee: a prospective cohort study.

Authors:
Thomas Benz (t.benz@rehaclinic.ch)
Felix Angst (f.angst@rehaclinic.ch)
Susanne Lehmann (s.lehmann@rehaclinic.ch)
André Aeschlimann (a.aeschlimann@rehaclinic.ch)

Version: 2 Date: 30. April 2013

Author’s response to reviews: see over
Reviewer's report

Title: Association of the sense of coherence with physical and psychosocial health in the rehabilitation of osteoarthritis of the hip and knee: a prospective cohort study.

Version: 2 Date: 11 April 2013

Reviewer: Christine Cedraschi

Reviewer's report:
I am satisfied with the authors' responses to my queries

- Review #1 has no suggested changes for the manuscript.

Level of interest: An article whose findings are important to those with closely related research interests

Quality of written English: Acceptable

Statistical review: No, the manuscript does not need to be seen by a statistician.

Declaration of competing interests:
I declare that I have no competing interests

Editorial Requirements:

1.) Name of ethics committee:
Please update your ethics statement to include the name of the ethics committee that approved your study.

- The ethics statement has been completed in Methods, Patients on page 7, end of paragraph: “The study’s protocol was approved by the Local Research Ethics Committee of the Health Department in Aarau, canton Aargau, Switzerland (EK AG 2008/026) and written informed consent was obtained from all participants.”